Author's response to reviews

Title: Post traumatic stress disorder and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study

Authors:

Shirley Telles (shirleytelles@gmail.com)
Meesha Joshi (meeshie26@gmail.com)

Version: 2 Date: 29 August 2009

Author's response to reviews:

Dear Sir or Madam,

In response to the letter from the Editorial board we have made the following changes:

1. Full details of all four authors have been added
2. The trial registration number has been added to the end of the abstract (please note that this a temporary number assigned by the Clinical Trials Registry of India). We expect to get the permanent number within the next 15 days.

Thanking you,
Sincerely,
Shirley Telles MBBS, Ph.D.